Case report

Anti-hormonal Therapy in the Treatment of Ductal Carcinoma of the Parotid Gland

Saul Campos-Gomez, Jose H Flores-Arredondo, Rita Dorantes-Heredia, Monica Chapa-Ibargüengoitia and Roberto De la Peña-Lopez

BMC Cancer

Ms. Patricia Marie Ratin
Journal Editorial Office
BioMed Central
E:editorial@biomedcentral.com
Thank you for your review of our manuscript. We have read your comments. Below, I would like to outline changes made:

1) On abstract section and case presentation we clarified the presence of parotid lesions and metastatic scalp lesions since initial presentation.

2) We stated the aim of the report in the last sentence of background.

3) We reviewed and added more recent reports of genetic profiling and target therapy on salivary ductal carcinoma

4) A pre, post treatment and postsurgical TC image are included on figure 1

5) A native English speaker has revised the text.

6) Haematoxylin-eosin staining, Her2, Estrogen and Progesterone receptor immunostains are described on case presentation and images included on figure 2

Again, thank you for your interest in our work. We hope to hear from the reviewers soon. Please do not hesitate to contact us with any questions you may have

Best Regards,

Saul Campos-Gomez, MD
Associate Professor, Medical Oncology
Centro Oncológico Estatal, ISSEMYM
Av. Solidaridad Las Torres No. 101
Col. Del Parque
Toluca, México CP 50180